Author's response to reviews

Title: Anomalous left coronary artery from the pulmonary artery (ALCAPA) presenting with sudden death in an octogenarian: A case report

Authors:

    ahmad Separham (a_separham2005@yahoo.com)
    Parvaneh Aliakbarzadeh (Aliakbarzadehp@yahoo.com)

Version: 8 Date: 28 October 2011

Author's response to reviews: see over
Anomalous left coronary artery from the pulmonary artery (ALCAPA) presenting with sudden death in an octogenarian: A case report

Ahmad Separham, MD, Interventional Cardiologist, Department of cardiology, Cardiovascular Research center, SHAHID MADANI HEART Hospital, Tabriz, Iran

Parvaneh Aliakbarzadeh, MD, Cardiologist, Department of cardiology, Cardiovascular Research center, SHAHID MADANI HEART Hospital, Tabriz, Iran

Corresponding author: Ahmad Separham, MD, Interventional Cardiologist, Department of cardiology, Cardiovascular Research center, SHAHID MADANI HEART Hospital, Tabriz, Iran
Tel: +984113363880 fax: +984113363880 E-mail address: a_separham2005@yahoo.com